Author's response to reviews

Title: Recall of intestinal helminthiasis by HIV-infected South Africans and misinterpretation of egg excretion in worm/HIV co-infection analysis: a questionnaire-based study

Authors:

Vera VJ Adams (vera.arendse@mrc.ac.za)
Miles MB Markus (medsynth@yahoo.co.uk)
Zilungile ZL Kwitshana (zilungile.kwitshana@mrc.ac.za)
Muhammad MA Dhansay (ali.dhansay@mrc.ac.za)
Lize L van der Merwe (lize.van.der.merwe@mrc.ac.za)
Gerhard G Walzl (gwalzl@sun.ac.za)
John JE Fincham (john.finchem@mrc.ac.za)

Version: 5 Date: 24 April 2006

Author's response to reviews:

Reference 42 was given as "in press". Complete publication detail is now provided.